Author's response to reviews

Title: Prevalence and factors associated with Posttraumatic Stress Disorder seven years after the conflict in three districts in northern Uganda (The Wayo-Nero Study)

Authors:

    James Mugisha (jmmugi77@hotmail.com)
    Herbert Muyinda (muyindaherbert@hotmail.com)
    Peter Wandiembe (symon.wand@gmail.com)
    Eugene Kinyanda (Eugene.Kinyanda@mrcuganda.org)

Version: 8 Date: 18 June 2015

Author's response to reviews: see over
Comment:
One minor comment: in Table 2 the authors should include the
cutoffs or percentiles of what "low, medium, and high" means in
the childhood trauma index. The same, if possible, for the
Negative event coping style index.

Response
Now included (percentiles in the table)